Author’s response to reviews

Title: Disease characteristics and treatment patterns of Chinese patients with colorectal cancer: a retrospective study using medical records from China

Authors:

Ruihua Xu (ruihxu@163.com)
Wei Wang (wangwei@fsyyyy.com)
Bo Zhu (bo.zhu@tmmu.edu.cn)
Xiaoyan Lin (xiaoyanlin@yahoo.com)
Dong Ma (Madong8005@126.com)
Lingjun Zhu (zhulingjun@njmu.edu.cn)
Qingchuan Zhao (qingchuangzhao@163.com)
Yongzhan Nie (niyongzhan@qq.com)
Xiaohong Cai (xiaohong_cai@hotmail.com)
Qi Li (Leeqi2001@hotmail.com)
Weijia Fang (weijiafang@zju.edu.cn)
Hongyan Li (li_hong_yan_sh@lilly.com)
Ning Wang (wang_ning4@lilly.com)
Yun Chen (chen_yun_kathy@lilly.com)
Cike Peng (Peng_ci_ke@lilly.com)
Honghao Fang (honghao.fang@iqvia.com)
Lin Shen (linshenpku@163.com)

Version: 1 Date: 12 Aug 2019

Author’s response to reviews:

As suggested by the editor, we corrected the manuscript and submitted it again.